Author’s response to reviews

Title: Hallux valgus and plantar pressure loading: the Framingham Foot Study

Authors:

Andrew M Galica (andygalica@gmail.com)
Thomas J Hagedorn (tommyhagedorn@gmail.com)
Alyssa B Dufour (alyssadufour@hsl.harvard.edu)
Jody L Riskowski (jriskowski@gmail.com)
Howard J Hillstrom (hhillstrom@hss.edu)
Virginia A Casey (virginiacasey@hsl.harvard.edu)
Marian T Hannan (hannan@hsl.harvard.edu)

Version: 4 Date: 10 October 2013

Author’s response to reviews:

Dear Editor:

The capital V has been removed from the title in the manuscript and the online submission.

Figure 2: This figure is similar to the Figure 1. in Riskowski et al which will appear in the November 2013 issue of Arthritis Care Res. We are in the process of getting permission from the publisher of this journal (Wiley Publishing). We do not anticipate any problems with this and will pass along the final word once we have it.

thank you.